Author’s response to reviews

Title: The necessity and optimal time for performing pars plana vitrectomy in acute retinal necrosis patients

Authors:

Shulin Liu (shulinliu_cmu@163.com)
Desai Wang (491071939@qq.com)
Xuedong Zhang (zxued@sina.com)

Version: 2 Date: 02 Jan 2018

Author’s response to reviews:

Dear Editors and Reviewers:

Thank you for your letter and for the reviewers’ comments concerning our manuscript entitled “The necessity and optimal time for performing pars plana vitrectomy in acute retinal necrosis patients” (ID: BOPH-D-17-00534R1). Those comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our researches. We have studied comments carefully and have made correction which we hope meet with approval.

Yours sincerely,

Xuedong Zhang